%% Math 131, Fall, 2003
\documentclass[12pt]{article}
\usepackage{latexsym,amsmath,epic,eepic,pstcol}
\usepackage{math131}

\pagestyle{empty}

\begin{document}

\noindent\textbf{Name}:\ \vbox{\hrule width 3in}

\begin{center}
\textbf{Holy Cross College, Fall Semester, 2003} \\
\textbf{Math 131, Quiz \#9, Friday, November 14} \\
Professor Hwang \\
\end{center}

\be

\item Let $f(x)=x^3-12x+3$. Find the critical and inflection point(s)
of~$f$, and classify each critical point as a local maximum, local
minimum, or neither. (Hint: Make a sign diagram for~$f'$.)

\vfil\vfil

\item Find the absolute maximum and absolute minimum values of
  $f(x)=x^3-12x+3$ on the interval $[-1,3]$. (Note: This is the same
  function as in the previous question.)

\vfil

\ee
\eject

\end{document}
